Author’s response to reviews

Title: Risk factors for failure of revision total hip arthroplasty using a Kerboull-type acetabular reinforcement device

Authors:
Shinya Hayashi (s11793290@yahoo.co.jp)
Takayuki Nishiyama (takakyon2@aol.com)
Shingo Hashimoto (shingo1023@theia.ocn.ne.jp)
Tomoyuki Matsumoto (matsun@m4.dion.ne.jp)
Koji Takayama (kojitakayama1978@gmail.com)
Kazunari Ishida (s11793290ishida@yahoo.co.jp)
Kotaro Nishida (s11793290nishida@yahoo.co.jp)
Ryosuke Kuroda (s11793290kuroda@yahoo.co.jp)

Version: 3 Date: 01 Aug 2017

Author’s response to reviews:

Answer to Editor comments: Editor Comments: Figure 2: The femoral component (stem) is still not visible. It is a quality-control to display the entire prosthesis. Answer: Thank you for your suggestion. We changed the photograph to display the entire prosthesis. Figure 6: Please change the label of the y-axis to survival rate (%). Answer: Thank you for your suggestion. We changed the label of the y-axis to survival rate (%).